\documentclass[11pt]{article}

\usepackage{fullpage,times,graphicx}

\frenchspacing

\long\def\comment#1{}

\begin{document}

\noindent
Name \rule{3in}{0.01in}\\[0.5in]
Email \rule{3in}{0.01in}

\begin{center}
\Large Reading Quiz for Sheather Chapter 6
\end{center}

~\\[3em]

\begin{enumerate}

\item Download the file ``pgatour2006.csv'' from Canvas, week05 folder in the files area, and read it into R.

There are 11 numerical variables in this data set.  Which variables need to be transformed so that they are more nearly normally distributed (or at least symmetric \& unimodal)?  Fill out the following table\footnote{The variable ``PrizeMoney'' is the average prize money per tournament  for each player.  The other variabels are features of the tournament and/or each player's performance.}:

\long\def\answers{
\begin{itemize}
\item \mbox{\hbox to 1.6in{\hfill Transform? (yes/no)}}  \rule{4in}{.01in}\rule{0in}{.5in}
\item \mbox{\hbox to 1.6in{\hfill Justify your answer:}} \rule{4in}{.01in}\rule{0in}{.5in}
\item \mbox{\hbox to 1.6in{\hfill If yes, what transformation?}} \rule{4in}{.01in}\rule{0in}{.5in}\\
\end{itemize}
}

\begin{itemize}
\item TigerWoods % 1 or 0, whether Tiger Woods played in the tournament)
\answers
\item PrizeMoney (average prize moneypar per tournament)
\answers
\item AveDrivingDistance % (average distance to hit ball from the ``tee''\footnote{The goal of golf is to get the ball from the ``tee'' to the ``hole'' in as few hits as possible. The first ``hit'' is called a ``tee shot'', also called a ``drive''.})
\answers
\item DrivingAccuracy % (percent of time player is able to hit the fairway\footnote{Well-maintained part of course between ``tee'' and ``hole''.} with tee shot)
\answers
\item GIR % (percent of time player is able to hit the ``green'' in with at least two shots to spare to make ``par''\footnote{``par'' is the expected number of hits, from ``tee'' to ``hold''.}, to hit the ``hole'')
\answers
\item PuttingAverage % (average number of hits to reach the ``hole'' if the ball has been hit GIR.)
\answers
\item BirdieConversion % (percent of time player makes it to the hole in one hit under ``par'', after hitting GIR)item SandSaves (Percent of time player was able to get free of a sand trap)
\answers
\item Scrambling % (percent of time player does not make GIR, but makes hole in par or less)
\answers
\item BounceBack % (???)
\answers
\item PuttsPerRound %  (average total number of putts per round)
\answers
\end{itemize}

\end{enumerate}

\end{document}
